\documentclass{simiode}
% SIMIODETemplate-TeacherORStudent.tex and simiode.cls for LaTeX2e
% Version 29 July 2015.
%
% This class (simiode.cls) is a modification of that used by
% the journal PRIMUS - Problems, Resources, and Issues in
% Mathematics Undergraduate Studies.
%
% See http://www.tandfonline.com/loi/upri .
%
% The class was developed by  Marc Frantz with suggestions of the
% PRIMUS Editorial Board, especially John Thoo.
%
% This template is to be used with the  class file "simiode.cls".
%
% The latest version of this class and template files
% can be found at
%
%       <http://www.SIMIODE.org> in the Author section.
%
% A "Quick Start Guide" appears below.
%
% This file must be processed with version LaTeX2e or higher.
%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% With this version of the tex file and the associated SIMIODE.cls file you can produce BOTH a Student Version and a Teacher Version of your modeling scenario or technique narrative.
%
% Basically the Student Version contains the title STUDENT VERSION along with the NAME OF SCENARIO, with no abstract, no keywords, no tags, no author identifiers, and no COMMENTS section. The Student Version contains only the STATEMENT of the modeling scenario with appropriate figures, data, instructions, and perhaps a bibliography appropriate for student resourcing, but nothing in the bibliography which would "give away" the nature of the modeling scenario, etc.
%
%The Teacher Version contains the material in the Student Version, namely the title NAME OF SCENARIO, now headed with   TEACHER VERSION, but also full author  identification, abstract, key words, tag words, and, following the STATEMENT, a section of COMMENTS which can be discussion of presentation  ideas, "solutions," techniques, etc.
%
% Authors can prepare a full Teacher Version using the notions immediately below  to create a pdf for review (but in review please eliminate all information about the author as we use a double-blind referee system. Such information will be restored when accepted and published in SIMIODE at www.simiode.org.
%
%Toggle the following line to turn the teacher version on (1) or off (0).
\def\TeacherEdition{0} % Toggle this line's value.
\newcommand{\teacher}[1]{\ifnum\TeacherEdition=1 #1 \fi}
% The \teacher{ } command wraps any text that will appear only in the teacher version of
% the document. See this file for examples. This has been done for you in the document below.
% Rename the pdf after you compile each final version.
% This option was created  by Eric Sullivan, Carroll College, Helena MT.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\usepackage{amsmath,amssymb}
\usepackage{color}
\usepackage{graphicx}
\usepackage{hyperref}
\usepackage{subfigure}



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% NOTE regarding double spacing %%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Manuscripts should be submitted doubled spaced.  This is
% accomplished via the \baselinestretch command below.
% However, you may obtain an idea of how your article will
% look in final form, so that you can better plan your
% presentation.  To see your manuscript formatted very
% similarly to the on-line published format of  SIMIODE,
% comment the following line out by inserting a percent sign % (%) at its beginning.

\renewcommand{\baselinestretch}{1.3}

%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% NOTE regarding macros %%
%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Any macros defined for your paper should be contained in
% the top matter. Likewise, any environment definitions such
% as \newtheorem or \newenvironment should also go in the
% top matter.
%
% Do not \input your macros as separate files within your
% final version because more files make it harder for the
% editors and users to keep track of and/or modify and
% append your paper.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Top matter is above and
% the beginning of the document below.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{document}
\begin{thebibliography}{99}

\bibitem{cc2003}Cantrell, R.S.  and C.~Cosner. 2003. {\it Spatial Ecology via Reaction-Diffusion Equations\/}. Chichester, West Sussex, UK: John Wiley \& Sons.

\bibitem{clark2010}  Clark, C. W. 1990. {\it Mathematical bioeconomics: The optimal management of renewable resources, 2nd edition\/}.  New York: John Wiley \& Sons.

\bibitem{ding2009} Ding, W.  and S.~Lenhart. 2009.
Optimal Harvesting of a Spatially Explicit Fishery Model. {\it Natural Resource Modeling\/}.  22:(2):  173-211.

\bibitem{joshi2009} Joshi, H.R., G.E.~Herrera, S.~Lenhart, and M.G.~Neubert. 2009 Optimal Dynamic Harvest of a Mobile Renewable Resource.
{\it Natural Resource Modeling\/}.  22(2): 322-343.

\bibitem{hilborn1992} Hilborn, R.  and C.~J.~Walters. 1992. {\it Quantitative Fisheries Stock Assessment:  Choice,
Dynamics and Uncertainty\/}. London:  Chapman and Hall, Inc.

\bibitem{hilborn2012} Hilborn, R.  2012.   {\it Overfishing:   What  Everyone  Needs  to  Know\/}.  Oxford: Oxford  University
Press.

\bibitem{k1998} Kurlansky, M. 1998.    {\it Cod:  A  biography  of  the  fish  that  changed  the  world\/}.  London: Penguin
Books.



\bibitem{nolen} Nolen, J. 2009.  {\it Partial Differential Equations and Diffusion Processes\/}. \href{http://math.stanford.edu/~ryzhik/STANFORD/STANF227-12/notes227-09.pdf}{http://math.stanford.edu/~ryzhik/STANFORD/STANF227-12/notes227-09.pdf}. Accessed 8 December 2015.


\bibitem{quinn2003} Quinn, T.J.   II. 2003. Ruminations on the Development and Future of Population
Dynamics Models in Fisheries. {\it Natural Resource Modeling\/}. 16(4): 341-392, 2003.

\bibitem{uscod} Risky Decisions: How denial and delay brought disaster to New England's historic fishing grounds: A brief from The PEW Charitable Trusts. October, 2014. 
\href{http://www.pewtrusts.org/\~/media/assets/2014/09/risky-decisions-brief\_final.pdf?la=en}{http://www.pewtrusts.org/\~/media/assets/2014/09/risky-decisions-brief\_final.pdf?la=en}.  Accessed 8 December 2015.

\bibitem{yakubu2011} Yakubu, A.A.,  N.~Li, J.M.~Conrad, and M-L.~Zeeman. 2011. Constant proportion harvest policies: dynamic implications in the Pacific halibut and Atlantic cod fisheries. {\it Mathematical Biosciences\/}. 232: 66-77.

\end{thebibliography}


\newpage
% You only need to change the NAME OF SCENARIO in the next line and delete the footnote.
\title{\ifnum\TeacherEdition=1 TEACHER VERSION  \else STUDENT VERSION \fi\\FISHERY HARVESTING: ATLANTIC COD}






 \markboth{Fishery Harvesting}{Fishery Harvesting}






\author{Wandi Ding \\
Middle Tennessee State University \\
Murfreesboro  TN 37132 USA  \\
Wandi.Ding@mtsu.edu
}
\makeStitlePDFLaTex


\begin{abstract}
We offer students a modeling opportunity with logistic growth, harvesting, and diffusion in analyzing ocean fisheries which was adapted from \cite{ding2009, joshi2009}, in particular, that of the Atlantic cod . We help students build models, ever more complex, to capture physical realities. At each stage we ask students to reflect on the model and the implications.
\end{abstract}




\section*{SCENARIO DESCRIPTION}

The logistic differential equation is often used to model the spread  of  disease as well as population growth. Can we use it to model sustainable harvesting of natural resources?

Fish are a valuable source of protein, and many people have lived to a large extent on fish
and other seafood.  Over fishing has driven many stocks of fish to near extinction \cite{hilborn1992, hilborn2012}.
This applies in particular to the Mediterranean Sea, the Baltic Sea, and the North Atlantic
Ocean.  The collapse of the Newfoundland or Baltic sea cod fisheries should be taken as a
severe warning that the fishery industry needs more careful controls \cite{k1998}.  With appropriate
stock  assessment  data,  mathematical  models  can  be  used  to  derive  possible  management
strategies, which may aid the supervision of this industry.

U.S. stocks of Atlantic cod came close to commercial collapse in the mid-1990s. The 2012 assessments of Gulf of Maine and Georges Bank cod indicated both stocks are seriously over fished and are not recovering as quickly as expected. Based on these assessments, quotas for both stocks were significantly reduced in 2013 to help ensure over fishing does not occur and that these stocks rebuild. The Gulf of Maine cod quota was cut by 80\% , and the Georges Bank cod quota was cut by 61\%. National Oceanic and Atmospheric Administration (NOAA) Fisheries and the New England Fishery Management Council continue to work on management measures that will further protect cod stocks and provide opportunities for fishermen to target other healthy fish stocks instead of cod \cite{uscod}. Now NOAA asks you to model the fish stock with harvesting in St. Georges Bank in order to fish sustainably.

\begin{figure}
 \centering
{ \includegraphics[width=0.8\textwidth]{./1-70-FisheryHarvestFigure}}
   \caption{Atlantic Cod Biomass }
\end{figure}



\section*{Part I: Modeling Fish Harvesting}
Let us consider the cod existing in a continuous, finite
region with logistic growth and carrying capacity $K$, where the harvesting of cod is proportional to the stock available.

\begin{itemize}
\item[1.] Define your variables and write down the differential equation for the Atlantic cod with harvesting proportional to the stock available.
Suppose $u(t)$ is the population of cod at time $t$, $r$ is the growth rate of the cod, $h$ is the constant harvesting rate, $u_0$ is the initial cod population. Then a reasonable model \cite{clark2010} is given by:

 \begin{align}\label{equation:logistic}
&\frac{du}{dt} = r \, u\left(1-\frac{u}{K}\right) - h \,  u , \quad  0 < t < T, \\
& u(0) = u_0, \nonumber
\end{align}
\noindent where $T$ is the length of the fishing period.

\item[2.] Before solving the differential equation (\ref{equation:logistic}), what do you expect of the behavior of the fish stock? E.g., is the fish stock increasing or decreasing?

\item[3.] What kind of differential equation is this? Is it linear or nonlinear? What kind of method might you use to solve it?


\vspace*{-.3 true cm}
\begin{table}[h]
\begin{center}
\begin{tabular}{llllll}
Year & $X_t$ & $h_t$ & Year & $X_t$ & $h_t$ \\ \hline

1978 & 72,148 & 0.18847 & 1994 & 21,980 & 0.282701\\

1979 & 73,793 & 0.149741 & 1995 & 17,463 & 0.199275 \\

1980 & 74,082 & 0.219209 & 1996 & 18,057 & 0.18781 \\

1981 & 92,912 & 0.176781 & 1997 & 22,681 & 0.193574 \\

1982 & 82,323 & 0.282033 & 1998 & 20,196 & 0.189526 \\

1983 & 59,073 & 0.34528 & 1999 & 25,776 & 0.170108 \\

1984 & 59,920 & 0.206545 & 2000 & 23,796 & 0.156601 \\

1985 & 48,789 & 0.338185 & 2001 & 19,240 & 0.281787 \\

1986 & 70,638 & 0.147236 & 2002 & 16,495 & 0.252869 \\

1987 & 67,462 & 0.19757 & 2003 & 12,167 & 0.255417 \\

1988 & 68,702 & 0.231541 & 2004 & 21,104 & 0.081034 \\

1989 & 61,191 & 0.208597 & 2005 & 18,871 & 0.0873972 \\

1990 & 49,599 & 0.335648 & 2006 & 21,241 & 0.0819517 \\

1991 & 46,266 & 0.295344 & 2007 & 22,962 & 0.105181 \\

1992  & 34,877 & 0.331848 & 2008 & 21,848 & unknown \\

1993 & 28,827 & 0.350394 & 2009 & - & - \\ \hline
\end{tabular}
\parbox{10.5 true cm}{\caption{\label{table:FishData} Annual (1978-2008) values of Atlantic cod biomass in metric tons, $X_t$, and harvest rate, $h(t)$, in metric tons in Georges Bank from \cite{yakubu2011}.}}
\end{center}
\end{table}

\vspace*{-.5 true cm}
\item[4.] Find the solution of the model. Atlantic cod biomass (in metric tons) and harvest rate in Georges Bank \cite{yakubu2011} are given in Table 1 from 1978 to 2008,  could you estimate the parameters $r$ and $K$ from the data?

\item[5.] What is the long term behavior of the cod population? What does harvesting do for the cod? Can we conserve the cod population?
Using the data  in Table 1,  if we increase the constant harvest rate, $h,$ to be 0.4, how will the population of Atlantic Cod change over time?
\item[6.] Constant harvesting may not be realistic. What other harvesting strategy can you think of? Address the issues (1) - (5) above for this harvesting strategy.

\end{itemize}


\section*{Part II: Challenge: Farm Fishing}
There are several recent issues of {\it Natural Resource Modeling Journal\/} devoted
to fishery management and we call attention to the survey paper \cite{quinn2003}. The author  traces the development of fisheries
models from 1900 to the $21^{\rm st}$ century. He points out the fact that future
fishery models will need to  better address  habitat and spatial
concerns and  to understand the effects of harvesting on the ecosystem. Managers have begun to use spatial management instruments, in the form of both permanent and temporary closures.


Assuming we implement fish farming for cod, so that cod live in a bounded region, for simplicity, we assume one-dimensional spatial domain with logistic growth and continuous diffusion \cite{cc2003}. We assume zero stock at the ends of the habitat, depicting a situation in which the habitat everywhere outside the  domain in question cannot support the resource.

Let's think about how to add the spatial component. Hint to students: the term  $u_{xx}$ can describe the diffusion \cite{cc2003}. In the Appendix, you may find an intuitive way to derive and understand the diffusion partial differential equation. Note the students from Differential Equations II will be comfortable using $u_{xx}$, if this part is too challenging, you may skip questions 7-9.

\begin{itemize}
\item[7.] We write down the new differential equation that includes the spatial component, let $u=u(x,t).$

\begin{equation}\label{equation:diffusion}
u_t= D u_{xx} + r\, u\left(1-\frac{u}{K}\right) - h\, u , \quad  0 < x < l, \ 0<t<T
\end{equation}
What kind of initial and/or boundary conditions do we need to solve this problem? Hint:
We need to specify both initial condition $u(x,0) = f(x)$ and boundary conditions $u(0,t) = g_1(t)$ and $u(l,t) = g_2(t)$ (in our case, we have $g_1(t)=g_2(t)=0$ since we have hostile environment). Interpret these conditions in this fishery setting. $D$ is the diffusion rate.


\item[8.] How might a solution strategy differ from Part I used in solving (\ref{equation:logistic})?   What do you anticipate from your model by adding the spatial component?

\item[9.] {\bf Ponder:} For an even more realistic situation, suppose the harvesting $h$ depends on the spatial location and the time.  Write out a differential equation which might model this phenomenon. What is the difference between this newer result and your result from Question 8?

\end{itemize}



\subsection*{Acknowledgement} We used a simplified version of the models from  W. Ding, G.E.~Herrera, H.R.~Joshi, S.~Lenhart and M.G.~Neubert \cite{ding2009, joshi2009}.

\teacher{
\section*{NOTES FOR TEACHERS}

Students will feel  comfortable solving Part I and be able to answer most questions from (1) - (7). For Part II, our original idea was a motivation for the introduction to partial differential equations for Differential Equations II class and for a graduate Mathematical Modeling class for Computational Science Program. The object is to give the students an opportunity to see more realistic models on fishery harvesting by adding the spatial component. Some students already know the diffusion term $u_{xx},$ if the students do not know this, it would be easier to tell them directly to use second order derivative for the diffusion as \cite{cc2003} suggests and not to spend too much time discussing it.
}

\section*{Appendix: Diffusion}

Description of movement arising from many short movements in random directions. Suppose an organism moves along a line: moving a distance $\Delta x$ to the left with probability $\frac{1}{2},$ moving a distance $\Delta x$ to the right with probability $\frac{1}{2}.$ Suppose $p(x,t)=$ probability that the organism is at location $x$ at time $t,$ then

\begin{equation}
p(x,t)=\frac{1}{2}p(x+\Delta x, t-\Delta t)+ \frac{1}{2}p(x-\Delta x, t-\Delta t).
\end{equation}

\noindent Subtracting $p(x,t-\Delta t)$ from both sides and divided by $\Delta t,$ we have

\begin{equation}
\displaystyle\frac{p(x,t)-p(x,t-\Delta t)}{\Delta t} = \frac{p(x+\Delta x,t-\Delta t)-2p(x,t-\Delta t)+ p(x-\Delta x, t-\Delta t)}{2\Delta t}.
\end{equation}

\noindent Imposing the scaling $\frac{(\Delta x)^2}{\Delta t}=2D,$ we obtain

\begin{equation}
\displaystyle\frac{p(x,t)-p(x,t-\Delta t)}{\Delta t} =\displaystyle\frac{D\Big[ p(x+\Delta x, t-\Delta t) -2p(x,t-\Delta t) +p(x-\Delta x, t-\Delta t)  \Big]}{(\Delta x)^2}.
\end{equation}

\noindent As $\Delta x, \Delta t$ go to $0,$ $\displaystyle\frac{p(x,t)-p(x,t-\Delta t)}{\Delta t}$ goes to $p_t(x,t),$ and

\begin{equation*}
\frac{\Big[ p(x+\Delta x, t-\Delta t) -2p(x,t-\Delta t) +p(x-\Delta x, t-\Delta t)  \Big]}{(\Delta x)^2}
\end{equation*} 
goes to $p_{xx}(x,t).$ So we have the diffusion equation
$$p_t = Dp_{xx}.$$





\end{document}
